\begin{table}%t2
\caption {\label{table:periods}Rotation periods ($P$), false-alarm probabilities (FAP), and light-curve amplitudes ($\Delta V$, $\Delta R$, $\Delta I$, $\Delta J$, 
$\Delta H$, and $\Delta K'$) of our targets.  A colon marks periods with FAP~$>$~5\%.} 
%\centerline
 {
\begin{tabular}{l l l l l l l l l l l l l l} 
\hline\hline 
  Id   &   $P$    &   Err    & FAP &  Band & $\Delta V$ & $\Delta R$ & $\Delta I$ & $\Delta J$ & $\Delta H$ & $\Delta K'$ & $P_{\rm Stas}^{{a}}$ & $P_{\rm Reb}^{{b}}$ & $P_{\rm Carp}^{{c}}$ \\ 
       &  (days)  &  (days)  &	   &       &  (mag)     &  (mag)     &  (mag)	 &  (mag)     &    (mag)   &	(mag)	&  (days)     &   (days)  &   (days)     \\ 
\hline	     				     1   & 9.78$^{{c}}$ &   ...    &        ...	 &   ...    &	...    &   ...    &    ...   &    ...	 &    ...    &   ...	&   9.78$^{{\rm d}}$  &   ...  &   ...   \\ 
   2   &   7.751  &   0.108  & $3\times 10^{-5}$  &  $RIJH$  &	...    &  0.085   &  0.065   &   0.040   &   0.035   &  0.025	&   ...   &   ...  &   ...   \\ 
   3   &   1.311  &   0.003  & $1\times 10^{-10}$ &  $IHJ $  &	0.050  &   ...    &  0.040   &   0.030   &   0.030   &  0.020	&   ...   &   ...  &   ...   \\ 
   5   &   ...    &   ...    &        ...	 &   ...    &	...    &   ...    &    ...   &    ...	 &    ...    &   ...	&   ...   &   0.60 &   ...  \\ 
   6   &   5.993  &   0.070  & $6\times 10^{-3}$  &  $IRJH$  &	0.090  &  0.065   &  0.040   &   0.020   &   0.020   &  0.020	&   ...   &   ...  &   ...   \\ 
   9   &   0.5627 &   0.0002 & $1\times 10^{-11}$ &  $IJHR$  &	0.090  &  0.100   &  0.065   &   0.040   &   0.040   &  0.045	&   0.57  &   ...  &   ...   \\ 
  10   &   5.816  &   0.010  & $1\times 10^{-6}$  &  $RIHJ$  &	0.365  &  0.345   &  0.280   &   0.140   &   0.150   &  0.125	&   ...   &   5.90 &   5.88  \\  
  11   &   7.383  &   0.003  & $1\times 10^{-6}$  &  $HJIR$  &	0.850  &  0.970   &  1.150   &   1.250   &   1.180   &  0.900	&   ...   &   7.50 &   7.46  \\  
  14   &   4.047  &   0.003  & $5\times 10^{-10}$ &  $IHJR$  &	0.390  &  0.345   &  0.265   &   0.135   &   0.145   &  0.125	&   4.02  &   4.04 &   3.98  \\ 
  15   &  11.86:  &   0.08   & $9\times 10^{-2}$  &  $IJH $  &	...    &   ...    &  0.160   &   0.030   &   0.035   &  0.030	&   ...   &   ...  &   ...   \\ 
  16   &   6.351  &   0.072  & $2\times 10^{-2}$  &  $IRJH$  &	0.145  &  0.130   &  0.055   &   0.035   &   0.035   &  0.030	&   ...   &   ...  &   ...   \\ 
  17   &   8.435  &   0.020  & $1\times 10^{-4}$  &  $IHJR$  &	0.395  &  0.290   &  0.245   &   0.125   &   0.130   &  0.115	&   7.70  &   4.18 &   ...  \\ 
  18   &  12.461  &   0.069  & $1\times 10^{-7}$  &  $RIHJ$  &	0.285  &  0.225   &  0.110   &   0.055   &   0.055   &  0.035	&   ...   &   ...  &   ...   \\ 
  19   &   0.605  &   0.035  & $4\times 10^{-4}$  &  $JI  $  &	...    &   ...    &  0.060   &   0.030   &    ...    &   ...	&   ...   &   ...  &   ...   \\ 
  29   &   8.468  &   0.046  & $1\times 10^{-2}$  &  $IJR$   &	0.175  &  0.150   &  0.145   &   0.040   &   0.050   &   ...	&   ...   &   ...  &   ...   \\ 
  35   &  12.521  &   0.218  & $3\times 10^{-4}$  &  $HJ  $  &	...    &   ...    &    ...   &   0.070   &   0.070   &   ...	&   ...   &  12.41 &   ...  \\ 
  40   &   5.297  &   0.007  & $5\times 10^{-7}$  &  $HJ  $  &	...    &   ...    &    ...   &   0.275   &   0.270   &  0.250	&   ...   &   ...  &   ...   \\  
  41   &   6.783  &   0.040  & $2\times 10^{-2}$  &  $JHIR$  &	0.260: &  0.205   &  0.125   &   0.075   &   0.090   &  0.110	&   ...   &   ...  &   ...   \\ 
  47   &  10.02   &   0.21   & $5\times 10^{-2}$  &  $HJ  $  &	...    &   ...    &    ...   &   0.055   &   0.055   &   ...	&   9.60  &  10.08 &   ...   \\ 
  48   &   7.721  &   0.019  & $2\times 10^{-4}$  &  $IHJ $  &	...    &   ...    &  0.280   &   0.270   &   0.230   &   ...	&   ...   &   ...  &   ...   \\ 
  51   &   2.763  &   0.010  & $2\times 10^{-2}$  &  $IJH $  &	...    &   ...    &  0.040   &   0.025   &   0.025   &  0.025	&   ...   &   2.81 &   ...   \\ 
  52   &   1.696  &   0.004  & $1\times 10^{-2}$  &  $IJHR$  &	0.170  &  0.090   &  0.070   &   0.030   &   0.035   &  0.050	&   1.69  &   1.70 &   ...   \\ 
  54   &   2.514  &   0.002  & $5\times 10^{-12}$ &  $JHI $  &	...    &   ...    &  0.315   &   0.265   &   0.235   &  0.190	&   ...   &   ...  &   ...   \\ 
  55   &   8.432  &   0.035  & $5\times 10^{-6}$  &  $IRJH$  &	0.300: &  0.230   &  0.135   &   0.070   &   0.120   &   ...	&   8.06  &   ...  &   ...   \\ 
  56   &   3.80   &   0.06   & $3\times 10^{-2}$  &  $R$     &	...    &  0.065   &    ...   &    ...	 &    ...    &   ...	&   3.86  &   3.84 &   ...   \\ 
  57   &   4.647  &   0.038  & $1\times 10^{-2 }$ &  $JIR $  &	0.185  &  0.120   &  0.050   &   0.040   &    ...    &   ...	&   ...   &   ...  &   ...   \\ 
  59   &   ...    &   ...    &        ...	 &   ...    &	...    &   0.5    &    0.4   &    0.3	 &    ...    &   ...	&   7.33  &   7.33 &   ...   \\  
  60   &   ...    &   ...    &        ...	 &   ...    &	...    &   ...    &    ...   &    ...	 &    ...    &   ...	&   4.04  &   4.11 &   4.09  \\ 
  61   &   ...    &   ...    &        ...	 &   ...    &	...    &   1.5    &    0.8   &    0.6	 &    ...    &   ...	&   ...   &  18.68 &   ...   \\ 
  62   &   ...    &   ...    &        ...	 &   ...    &	...    &   ...    &    ...   &    ...	 &    ...    &   ...	&   2.05  &   2.05 &   ...   \\ 
  63   &   ...    &   ...    &        ...	 &   ...    &	...    &   ...    &    ...   &    ...	 &    ...    &   ...	&   1.10  &   1.09 &   ...   \\ 
  66   &  21.68:  &   0.30   & $1\times 10^{-1}$  &  $HJ  $  &	...    &   ...    &    ...   &   0.40	 &   0.40    &   ...	&   ...   &  59.87 &  14.62  \\  
  69   &   6.550  &   0.068  & $5\times 10^{-3}$  &  $JH  $  &	...    &   ...    &    ...   &   0.060   &   0.050   &   ...	&   ...   &   ...  &   ...   \\  
  75   &   ...    &   ...    &        ...        &   ...    &	...    &   ...    &    ...   &    ...    &    ...    &   ...	&   ...   &  10.71 &   ...   \\  
  90   &   4.376  &   0.013  & $1\times 10^{-5}$  &  $IJRH$  &	0.205  &  0.135   &  0.120   &   0.055   &   0.040   &   ...	&   ...   &   ...  &   ...   \\ 
  91   &  10.176  &   0.055  & $2\times 10^{-5}$  &  $JHIR$  &	...    &  0.070   &  0.080   &   0.090   &   0.095   &  0.070	&   ...   &   ...  &   ...   \\ 
  93   &   5.70:  &   0.12   & $2\times 10^{-1}$  &  $IJ  $  &	...    &   ...    &  0.150   &   0.200   &    ...    &   ...	&   ...   &   5.53 &   5.53  \\  
\hline       	                 
\end{tabular}}			         	                	                 
 	\smallskip                 
$^{{a}}$   Rotational periods derived by \citet{stass99}. 
$^{{b}}$   Rotational periods derived by \citet{rebu01} or \citet{rebu06}. 
$^{{c}}$   Rotational periods derived by \citet{Carpenter01}. 
$^{{d}}$   Orbital period from \citet{stass06}, and confirmed by us.
\end{table}